Author's response to reviews

Title: Influence of uncorrected refractive error and unmet refractive error on visual impairment in a Brazilian population

Authors:

Fábio H Ferraz MD (ferrazhenrique@ig.com.br)
Jose’ E Corrente MD (jecorren@ibb.unesp.br)
Paula Opromolla MD (opromolla@saude.gov.br)
Silvana A Schellini Prof. (sartioli@fmb.unesp.br)

Version: 3
Date: 15 June 2014

Author's response to reviews: see over
MS: 164844033117437
Influence of uncorrected refractive error and unmet refractive error on visual impairment in a Brazilian population
Fábio H Ferraz MD, José E Corrente MD, Paula Opromolla MD and Silvana A Schellini Prof.

15/6/2014 from Perth (AU)

Dear Prof. Lambert,

We apologize for do not revised the article format.

We are sending now the revised manuscript, following the instructions for publication in BMC Ophthalmology:

1. all information about disclosures, competing interests and presentations from the title page were removed.
2. all the email addresses of all co-authors were included in the title page.
3. acknowledgements section after the authors' contributions section was changed and in accordance with the journal's instructions for authors
4. the tables were removed from the additional files and include these in the main manuscript after the references section.
5. Also we provided the abbreviations after the references.

I have one more concern about the section where the article was included. The subject match with epidemiology research but not Retina. Would you please tell me if is possible to change?

Silvana Schellini

Corresponding author